

The Australian Research Council Centre of Excellence for Children and Families over the Life Course Phone +61 7 3346 7477 Email lcc@uq.edu.au lifecoursecentre.org.au

WORKING PAPER SERIES

A DYADIC APPROACH TO THE STUDY OF PERCEIVED INFERTILITY AND CONTRACEPTIVE USE

Ester Lazzari

Australian National University

Edith Gray Australian National University

Bernard Baffour

Australian National University

No. 2021-13 July 2021















NON-TECHNICAL SUMMARY

There is an increasing literature on women's perception of infertility and contraceptive use, with studies suggesting that it is related to unintended pregnancy. Little research investigates the correlates of perceived infertility, and quantitative investigation of couple-level perceived infertility appears absent from the literature, which is somewhat surprising, as infertility is a couple-level outcome. Furthermore, studies that relate to perceived infertility and use of contraception, or lack thereof, are typically limited to young adults. The present study aims to answer the following two research questions: (a) Are the factors that affect the perception of infertility among couples gendered? and (b) To what extent the perception of infertility affects contraceptive use? Drawing from previous literature, the factors associated with the perception of infertility are grouped in two main categories: biological factors (such as age and perceived health status) and life-course interference factors (such as the desire and intention to have children, parity status and type of relationship). Using data sourced from the Household Income and Labour Dynamics in Australia (HILDA) survey, binary and multinomial logistic regression models are estimated to analyse the association between these two groups of variables with perceived infertility and the association of perceived infertility with contraceptive use among 1,654 couples. The results indicate that both biological and life-course interference factors are strong predictors of the perception of infertility at the couple level and that women's characteristics are more influential than their partners' characteristics in determining this perception. Additionally, couples with perceived infertility are less likely to use contraception, regardless of their short-term intention and desire to have children. This is the first paper to explore factors associated with perceived infertility using dyads rather than individuals as the unit of analysis and to provide a detailed analysis of an unexplored yet relevant reason why couples do not use contraception (the perception of infertility) among a nationally representative sample of couples.





ABOUT THE AUTHORS

Ester Lazzari holds an MSc in Economics and Social Sciences from Bocconi University in Milan (Italy) and she is a final year doctoral student in the School of Demography at the Australian National University. Her main research interests focus on the causes and consequences of late fertility, assisted reproductive technology and childlessness. Email: ester.lazzari@anu.edu.au

Edith Gray is a family demographer and Head of the School of Demography at the Australian National University. Edith specialises in the use of survey data to analyse fertility and family change and has made significant contributions to understanding contemporary patterns of Australian society. Her research has been published in leading journals including the Journal of Marriage and Family, Demography, and Population, Space and Place. Edith is a Council member of the International Union for the Scientific Study of Population, and member of a number of Government advisory committees. Email: edith.gray@anu.edu.au

Bernard Baffour is a Senior Lecturer at the School of Demography at the Australian National University. His primary research interest is survey methodology, including nonresponse, survey quality and small area estimation, applied to the analysis of complex data in understanding demographic processes. Email: bernard.baffour@anu.edu.au

DISCLAIMER: The content of this Working Paper does not necessarily reflect the views and opinions of the Life Course Centre. Responsibility for any information and views expressed in this Working Paper lies entirely with the author(s).





ABSTRACT

Using data sourced from the Household Income and Labour Dynamics in Australia (HILDA) survey, binary and multinomial logistic regression models are estimated to analyse the correlates of perceived infertility, and the relationship between perceived infertility and contraceptive use of 1,654 couples. Results show that both biological and life-course interference factors are strong predictors of the perception of infertility at the couple level, with women's characteristics more influential than their partner's characteristics. Additionally, couples with perceived infertility are less likely to use contraception, regardless of their short-term intentions or desire to have a child.

Keywords: Perceived infertility, life course, contraception, childbearing desires, dyads.

Suggested citation: Lazzari, E., Gray, E., & Baffour, B. (2021). 'A Dyadic Approach to the Study of Perceived Infertility and Contraceptive Use', Life Course Centre Working Paper Series, 2021-13. Institute for Social Science Research, The University of Queensland.





1. Introduction

There is an increasing literature on women's perception of infertility and contraceptive use, with many studies showing that it is related to unintended pregnancy (Gemmill 2018; Frohwirth, Moore & Maniaci 2013; Polis & Zabin 2012). Little research investigates the correlates of perceived infertility (Gemmill & Cowan 2021), and the quantitative investigation of couple-level perceived infertility appears absent from the literature, which is somewhat surprising, as infertility is a couple-level outcome. Some studies investigating the factors associated with perceived infertility, also included the perceived infertility of partners (Gemmill Sedlander & Bornstein 2020; Passet-Wittig *et al.* 2020), but due to data constraints they have not used the couple-dyad as the unit of analysis.

Perceived infertility, or low perceived susceptibility to pregnancy, is a major reason for unintended pregnancies occurring in the absence of contraception (Gemmill & Cowan 2021). The reason for this is that if individuals believe that they are at low risk of pregnancy, they may consider the use of contraceptives unnecessary. However, this provides a false sense of protection against unintended pregnancy for two main reasons. First, individual perceptions regarding their own infertility may not be accurate (Greil *et al.* 2014). Second, it has been shown that, even among couples with infertility, a natural conception can still occur after they cease trying to proactively conceive (Osmanagaoglu *et al.* 2002). This phenomenon is explained by the fact that infertility can be a temporary condition and not necessarily a permanent state. Most studies that relate to perceived infertility and use of contraception do not provide a comprehensive examination, as they are conducted with women who have unintended pregnancies and are often limited to young adults or university students (Polis and Zabin 2012; Gemmill 2018; Gemmill, Sedlande & Bornstein 2021).

In this paper, the correlates of perceived infertility are investigated and the association between perceived infertility and contraceptive use explored among a representative sample of Australian couples. The purpose is to understand the couple-level perceptions of infertility and their association with contraceptive use, specifically: (a) Are the factors that affect the perception of infertility among couples gendered? and, (b) To what extent does the perception of infertility affect contraceptive use? The data used to answer these questions are from the Household Income and Labour Dynamics in Australia (HILDA) survey, which provides a unique opportunity to investigate perceived infertility and contraceptive use as it contains reports from both members of a couple. This is the first paper to explore factors associated with perceived infertility using couple dyads rather than individuals as the unit of



analysis, and to investigate the relationship between perceived infertility and contraceptive use among a nationally representative sample.

2. Previous Research

Infertility is clinically defined as the failure to achieve a pregnancy after twelve months or more of regular and unprotected sexual intercourse (Zegers-Hochschild *et al.* 2017), whereas perceived infertility is a subjective measure that reflects individuals' beliefs regarding their own procreative ability and it is typically captured by social science surveys. Measures of perceived infertility and medical infertility are often not in agreement (Loftus 2009; Greil *et al.* 2010). Indeed, it was found that only about one in three women with infertility identifies as having a fertility problem (White *et al.* 2006), and that, by contrast, women may identify themselves as being infertile even if they do not meet the medical criteria for infertility (Polis & Zabin 2012).

Discrepancies between actual and perceived infertility are due to the existence of several non-medical factors that may affect people's belief regarding their own fecundity. The ways in which people make evaluations about their situation may vary as a function of their life-course goals (White *et al.* 2006). Previous studies drawing from the seminal work of Zola (1973) and Mechanic (1968) suggest that, at the individual level, symptom salience is an important predictor of perception of infertility (White *et al.* 2006) and of seeking help (Greil *et al.* 2013; Slauson-Blevins, McQuillan & Greil 2013). For women who wish to have a child, a lack of conception after unprotected sex is noticed and is often interpreted as a sign of infertility. In comparison women who have unprotected sex and who do not intend to become pregnant do not perceive infertility following the absence of pregnancy (Greil *et al.* 2010). In other words, infertility may not be perceived as a problem and it may even remain unnoticed, unless it interferes with individual fertility plans. The experience of infertility may be particularly distressing for women with no previous children (McQuillan *et al.* 2003; McQuillan, Stone & Greil 2007).

Relationship status is an important situational factor that can also affect one's ability to recognise the symptoms of infertility. Research has found that individuals are more likely to perceive infertility if they are in a relationship (Passet-Wittig *et al.* 2020; Polis *et al.* 2020; Gemmil & Cowan 2021) and that partnership stability and the partner's attitude toward childbearing can also affect one's ability to recognise the symptoms of infertility (Gemmil, Sedlander & Bornstein 2020; Passet-Wittig *et al.* 2020). These findings point at the inherently dyadic nature of the experience of infertility and confirm the importance of analysing infertility perceptions as a couple-level phenomenon. That the perception of



infertility is affected by life-course events is also evident by its instability over time (Passet-Wittig *et al.* 2020; Johnson *et al.* 2020).

Perceived infertility is a useful measure, as it reflects the meaning individuals make of their ability to reproduce, which provides a basis for understanding their fertility plans, intentions and behaviours (Shreffler *et al.* 2016; Johnson *et al.* 2020). Furthermore, the increasing literature on the relationship between susceptibility to pregnancy and contraceptive use illustrates its importance in the field of reproductive health. Several studies show that contraceptive use is more or less directly associated with the desire to have children: individuals in committed and long-term relationships manifest a lower desire to avoid pregnancy and higher fertility expectations (Barber *et al.* 2019; Wilson & Koo 2006; Weitzman *et al.* 2017) and, hence, are less likely to use contraception. However, the perception of low susceptibility to pregnancy may be another important mechanism explaining why women that do not intend to become pregnant cease using contraception (Gemmil 2018; Polis & Zabin 2012).

This paper builds on these previous two lines of research by investigating the association of couple's similarity and dissimilarity with respect to biological and life-course factors with perceived infertility, and, in turn, with contraceptive use. Data are sourced from the HILDA survey, which provides a unique opportunity to investigate perceived infertility and contraceptive use taking account of reports from both members of the couple. The data contains information that can inform the correlates of perceived infertility, as well as the relationship between perceived infertility and contraceptive use.

3. Theoretical Framework and Hypotheses

In this paper, the life-course approach (Elder 1994) is adopted as a framework through which to investigate perceived infertility and contraceptive use. In particular, we refer to an aspect that has received less attention in fertility research (Buhr & Huinink 2014; Settersten 2015): the core life-course principle of *linked lives*. Key transitions in life are linked to significant others: an individual's perception regarding the importance of becoming a parent and their expectation about when and if to have a child are influenced by the desires of other significant people, most importantly their partner.

Based on the literature review, the analysis focuses on two types of explanatory variables: biological factors and life-course interference factors, while also adjusting for standard socio-economic variables





and life satisfaction. Each of the two primary groups of factors has an individual-level and couple-level dimension, which is captured by measuring the degree of similarity or dissimilarity between partners. *Biological factors*. The investigation of the correlates of perceived infertility notably calls for the inclusion of the biological factors that affect fecundity. The ability to reproduce naturally declines with age. For instance, it has been estimated that only 75% of women trying to conceive at age 30 will have a conception ending in a birth within one year, 66% at age 35, and 44% at age 40 (Leridon 2004). Age is an important factor affecting the procreative ability of both men and women, although the decline in fecundity with age is slower for men (Liu & Case 2011; Schmidt *et al.* 2012).

Hypothesis 1: The couple's likelihood of perceiving infertility will be positively associated with an increase in both partners' age, but couples in which the female partner is older than the male partner will be more likely to perceive infertility compared to couples where the male partner is older.

Self-reported health status has been identified as being associated with a number of lifestyle risk factors that predispose individuals to infertility, such as smoking and extremes body mass index (BMI), and as having a strong association with infertility (Kelly-Weder & Cox 2006). The medical literature has shown that men and women are just as likely to contribute to the couple's fecundity (Brugh & Lipshultz 2004; Isidori *et al.* 2006), suggesting that couples in which at least one partner is in poor health status should be more likely to perceive infertility compared to couples in which both partners are in good health, regardless of whether the person in poor health is the male or female partner.

Hypothesis 2: The couple's likelihood of perceiving infertility will be equally associated with men's and women's self-rated health status.

Life-course interference factors. Infertility may remain unnoticed and not be perceived as a problem, unless it interferes with the couple's plans to have a child. In other words, if couples do not wish to have a child, can they experience infertility and would they label themselves as infertile? While infertility has a negative impact on the life of both men and women wishing to have a child, research has shown that within a couple, women are more likely than men to see infertility as a distressing experience (Greil, Leitko & Porter 1988; Hjelmstedt *et al.* 1999). The effect of a dissimilarity in childbearing desires on the perception of infertility has not been explored yet. In Australia, when couples disagree about wanting a child, women tend to be more influential than men in fertility decision-making (Testa & Bolano 2021),



which may suggest that women's fertility desires have a greater importance in determining couple perceptions.

Hypothesis 3: The strength of a couple's desire to bear future children will be positively associated with its perception of infertility. Additionally, if partners disagree about wishing a child, women's childbearing desires will have more influence on the perception of infertility than men's childbearing desires.

Further, research suggests that a couple's reproductive history can be crucial for understanding the salience of infertility (Passet-Wittig & Greil 2021), with the inability to reproduce coming as a particularly distressing experience among couples with no previous children (McQuillan *et al.* 2003; McQuillan, Stone & Greil 2007).

Hypothesis 4: A couple with one or more children will be less likely to perceive infertility compared to a couple with no children.

The interference of infertility with personal plans may be higher among married couples. Indeed, despite the increase in cohabitation in Australia, the majority of childbearing still happens within marriage (Australian Bureau of Statistics, 2017) and marriage is the transition most commonly associated with entry into parenthood (McDonald & Reimondos 2013).

Hypothesis 5: The likelihood of perceiving infertility will be higher among married couples than among cohabiting couples.

Contraceptive Use. There is a growing body of literature suggesting that low susceptibility to pregnancy is related to lower contraceptive use, and, in turn, to a higher risk of unintended pregnancies (Gemmill 2018; Gemmill Sedlander & Bornstein 2020; Frohwirth *et al.* 2013; Polis & Zabin 2012). A substantial proportion (57%) of unintended pregnancies in Australia occurs among couples that do not use birth control measures (Taft *et al.* 2018). Although there are several reasons why couples do not use contraception, in Australia (2012-2013), one study found that infertility of the woman or her partner was the most common reason for not using contraception among women in their reproductive ages not wanting a child (16-49) (Richters *et al.* 2016).





Hypothesis 6: The likelihood of using contraception will be lower among couples with a perception of infertility.

The following framework considers the factors associated with perceived infertility and contraceptive use.



Figure 1. Factors associated with the perception of infertility and contraceptive use among couples.

4. Data

This study uses wave 19 of the HILDA panel study (https://melbourneinstitute.unimelb. edu.au/hilda), collected in 2019 (see Watson & Wooden (2012) for information on the scope of the HILDA study). Wave 19 is the most recent wave that includes a special fertility module with questions on: contraceptive use, desires and intentions to have children, infertility, and sterility which can be used to understand fertility plans and perceptions taking account of reports from both members of the couple. The analytical sample consists of heterosexual couples where both partners are of reproductive age (women between 15 and 44 and men between 15 and 54) and in which the female partner was not pregnant at the time of the interview. Those that mentioned having had an operation that makes it impossible to have a child were also excluded (n=413). The final analytical sample interviewed totalled 1,654 couples, who were legally married or in a cohabiting relationship (Table 1).



Table 1. Selection of the Analytical Sample

	Couples in reproductive years*
Total in selected age range in 2019 wave	2,491
Exclusions:	
Female partner pregnant	85
One or both partners sterile	413
Missing data on dependent variable	339
Total selected	1,654
Of whom:	
Perceiving infertility	271
Not using contraception	390

Dependent variables.

The first dependent variable was constructed based on the following question on perceived infertility: *Based on medical advice, do you know of any physical or health reason that would make it difficult for you (and/or your partner) to have [children / more children]?* This question was not asked to people that provided a positive answer to the following question on permanent infertility (or sterility): *Have you ever had any operation that makes it impossible for you (and/or your partner) to have [a child / more] children?* as these respondents were excluded. These questions distinguish between couples who perceive infertility, where at least one partner answered yes to the first question, and couples where it is impossible to conceive. We note that the prevalence of perceived infertility may be underestimated because the question asks whether the perception of the respondent is based on medical advice, and so it may exclude individuals that perceive that they are infertile but who did not consult a physician. Perceived infertility is a dichotomous variable, coded as 1 if at least one partner perceived use: *Do you (and your partner) use birth control measures? (That is, are you using some form of contraception, including natural methods such as the rhythm method?)*. Contraceptive use is also a dichotomous variable, coded as 1 if at least one partner second as 1 if at least one partner uses contraception.

Independent variables.

The explanatory variables included three biological factors (age, age difference between partners, and perceived health status) and three life-course interference factors (desire for children, parity, and type of relationship).





Age. This variable is based on the age group of the female partner. In a couple, the male and female partner's ages are highly correlated. Hence, including both variables in the model would lead to unreliable regression coefficients. The regressions were repeated using the male partner's age and no significant difference was found in the results (not shown).

Age Difference between Partners. This variable captures potential gender differences in the relationship between age and perceived infertility. Three categories are considered: the partners of the couple are three or less years of age apart, the female partner is three or more years older than the male partner, or the male partner is three or more years older than the female partner.

Perceived Health Status. The variable on perceived health status is computed by combining both partners' perceptions of their health status. The variable is based on a question asking respondents to rate their health status as: excellent, very good, good, fair, or poor. Respondents answering that their health status was excellent, very good, or good are considered to be in good health, while respondents answering any of the other two options are considered to be in poor health. Four categories are considered: both partners are in good health, the female partner but not the male partner is in good health, the male partner but not the female partner is in good health, or neither of the partners is in good health.

Parity. This variable is based on the number of children ever born to the female partner. Couples are counted as childless if the female partner does not have any children, of first parity if the female partner has only one child, and of parity two or above if the female partner has two or more children. Parity information is based on the female partner, as men tend to report the occurrence of births with more errors compared to women (Rendall et al. 1999). A robustness check (not shown) revealed that there was no significant difference in the findings using information based on the male partner's parity history, as in only a small proportion of couples (8%) partners reported a different number of children ever born.

Desire for Children. Couple desire about having a child is computed by combining both partners desires to have children. The variable is derived from a question asking respondents to express their desire for one (more) child by choosing a number between 0 and 10, where a higher number indicates a stronger desire to have a child. A value above 4 has been considered as a sign of a strong desire to have a child. Different cut-points may also be used, however they cannot importantly affect the results, since fertility





desires tend to be polarized between those who definitely do not want to have a child and those who definitely want to have one (Wagner, Huinink & Liefbroer 2019). Such polarization can also be observed in the HILDA sample. Four categories are considered: neither of the partners wishes to have a child, the female partner but not the male partner wishes to have a child, the male partner but not the female partner wishes to have a child, or both partners wish to have a child.

Type of Relationship. The last predictor of interest is type of relationship, which indicates whether the couple is legally married or in a cohabitating relationship. This is a dichotomous variable, coded as 1 if the members of the couple are legally married.

In addition to the biological and life-course interference factors, control variables which are not of primary interest but that have been identified as influencing the perception of infertility or the use of contraception are also included.

Short-term Childbearing Intentions. To account for differences in motivation for pregnancy in the next twelve months, a measure of short-term childbearing intentions was added. This measure was grouped into four categories: neither of the partners has a short-term intention to have a child, the female partner but not the male partner has a short-term intention to have a child, the male partner but not the female partner has a short-term intention to have a child, the male partner but not the female partner has a short-term intention to have a child, or both partners have a short-term intention to have a child.

Life Satisfaction. Studies have shown that infertility is associated with lower levels of life satisfaction both at the individual and couple level (Klemetti et al. 2010; Luk & Loke 2015; McQuillan et al. 2007; Peterson, Newton & Rosen 2003). Additionally, life satisfaction might be an important confounding variable to control for as it is largely driven by personality traits (Schimmack *et al.* 2004). The variable is derived from a question asking respondents to express their satisfaction with their life by choosing a number between 0 and 10, where 0 indicates that the respondent is totally dissatisfied with life and 10 indicates that the respondent is totally satisfied with life. Since the variable is symmetrically distributed, it is dichotomised at the mean (Cohen 1983): coded as low if neither partner gave a score higher than 7.

Highest Level of Education. As previous research highlights that educational attainment is associated with contraceptive use (Frost, Singh & Finer 2007; Gemmill 2018) and with fertility knowledge and beliefs (Bunting, Tsibulsky & Boivin 2013; Gemmill & Cowan 2021), a two-category variable specifies whether both partners have a tertiary degree obtained through university.





Country of Birth. A large body of research, mainly from the United States, has shown that race has a well-established relationship with contraceptive use and fertility knowledge (Gemmill, Sedlander & Bornstein 2021; Yano, Lundsberg & Pal 2014). A four-category variable specifies whether both partners, the female partner only, the male partner only or neither partner were born in Australia or in another English speaking country.

Characteristics of both partners, and their similarity and dissimilarity, are analysed simultaneously in order to investigate gendered effects on the perception of infertility and contraceptive use. Table A1 in the Appendix shows the descriptive statistics for the measures used.

5. Methods

The analytical strategy employed to examine factors associated with both the perception of infertility and contraceptive use among couples is binomial logistic regression as both the dependent variables take the form of a discrete variable with two options. Then, to properly understand how perceived infertility and contraceptive use decisions of couples are influenced by life-course interference and biological factors, a multinomial logistic regression model was fitted. To do this, firstly a categorical variable was created from couples' answers to the perceived infertility and contraceptive use questions, with four values: 1 for couples that do not perceive infertility and use contraception, 2 for couples that do not perceive infertility and do not use contraception, 3 for couples that perceive infertility and use contraception, and 4 for couples that perceive infertility and do not use contraception. Since the response variable has no natural ordering, the subgroup of couples that do not perceive infertility and that use contraception was chosen as the reference category, as it corresponded to the highestnumbered sub-group (1,103). The fitted multinomial logistic model compares the reference to the remaining three subgroups, and investigates how the different life-course and biological factors affect couples' perceived fertility and contraceptive use simultaneously.

Partners were mostly concordant in their answers regarding the existence of a fertility problem and the non-utilization of contraception, with a disagreement rate of 8.5% and 9.5%, respectively. At all ages, women were more likely than men to perceive infertility, while the cases of disagreement were more symmetrically distributed across genders in the case of contraceptive non-use, with only slightly more women indicating that contraception was not used at ages 30-34 and above (Figure 2 and Figure 3). A sensitivity analysis sample was created, in which all cases of disagreement were excluded. Despite the p-





values being smaller due to the reduced sample size, coefficients maintained the same direction and remained statistically significant (not shown).



Figure 2. Congruence between partners' responses on perceived infertility, Australia, 2019. *Source*: Authors' calculations using data from the HILDA survey, wave 19, release 19.









6. Results

Correlates of perceived infertility

The results for the binomial regression analyses are presented in Table 2. Model 1 shows the estimated coefficients of the two main sets of explanatory variables: biological and life-course interference factors. Model 2, 3, and 4 include the control variables. A clear gradient with age is found, with higher odds of perceiving infertility for couples in which the woman is aged 40-44 (2.34). Couples in which the woman is more than three years older than the man also have higher odds of perceiving infertility (2.04), while no statistically significant effect is observed when the man is older. Couples in which both partners perceive to be in good health are not significantly different from couples in which only the woman is in good health in the perception of infertility; by contrast, couples in which the woman or both partners perceive to be in poor health have a higher likelihood of perceiving infertility (2.96 and 6.48, respectively). In Model 4, couples in which the woman only or both partners have a strong desire for children are more likely to perceive infertility (odds ratio 1.66 and 1.52, respectively), while if only the man has a strong desire for children, the effect on perceived infertility is not statistically significant. Compared to couples with two or more children, couples with one or no children are more likely to perceive infertility, although the magnitude of the association is larger for couples with only one child (2.97). Compared to marriage, cohabitation is largely protective against the perception of infertility (0.65). Results from Model 4 largely parallel those from Model 1, with few differences. First, the effect of perceived health status is reduced after including life satisfaction, although it remains statistically significant, indicating its independent effect on the perception of infertility. Second, the inclusion of educational attainment accentuates the effect of age on perceived infertility. This is explained by the fact that the biological clock plays a greater importance for highly educated men and women due to their higher tendency to delay childbearing (Lazzari 2021). However, due to the existence of a marked negative educational gradient with perceived infertility (Gemmill, Sedlander & Bornstein 2021), these two effects partly compensate for each other before controlling for educational attainment, leading to a weaker effect of age on perceived infertility.

Regarding the control variables, couples in which at least one partner was tertiary-educated had a lower odds of perceiving infertility (0.62) compared to couples where both partners had lower levels of education. Couples in which the woman or both partners were born overseas had lower odds of perceiving infertility (0.23 and 0.30, respectively) compared to the reference category. High life satisfaction was associated with a significantly lower likelihood of perceiving infertility (0.50).



	Perceived infertility (N=1,654) Odds ratios & 95% CI				
	Model 1	Model 2	Model 3	Model 4	
Biological factors					
Age ¹					
18-24 (ref.)	1.00	1.00	1.00	1.00	
25-29	1.26 (0.78, 2.06)	1.32 (0.81, 2.14)	1.37 (0.84, 2.24)	1.48 (0.90, 2.42)	
30-34	1.21 (0.72, 2.03)	1.24 (0.74, 2.10)	1.38 (0.82, 2.32)	1.47 (0.86, 2.49)	
35-39	1.71 (0.97, 3.01)	1.72 (0.98, 3.05)	2.03** (1.15, 4.61)	2.37** (1.33, 4.24)	
40-44	2.34** (1.28, 4.29)	2.36** (1.28, 4.34)	2.82*** (1.53, 5.21)	3.27*** (1.75, 6.13)	
Age difference					
No difference (ref.)					
W more than 3 years older	2.04** (1.18, 3.52)	2.08** (1.20, 3.61)	1.89* (1.09, 3.28)	1.88* (1.07, 3.29)	
M more than 3 years older	1.06 (0.78, 1.44)	1.07 (0.79, 1.46)	1.04 (0.77, 1.42)	1.14 (0.84, 1.57)	
Perceived health status					
Both good (ref.)	1.00	1.00	1.00	1.00	
W good, M poor	1.17 (0.70, 1.95)	1.00 (0.59, 1.69)	1.05 (0.62, 1.76)	0.86 (0.51, 1.47)	
W poor, M good	2.96*** (1.97, 4.44)	2.57*** (1.69, 3.91)	2.78*** (1.85, 4.19)	2.26*** (1.47, 3.48)	
Both poor	6.48*** (3.12, 13.41)	4.84*** (2.91, 12.55)	6.04*** (2.17, 9.89)	4.50*** (2.11, 9.62)	
Life-course interference					
factors					
Desire for children					
Both low (ref.)	1.00	1.00	1.00	1.00	
W high, M low	1.64 (0.96, 2.80)	1.61 (0.94, 2.75)	1.61 (0.94, 2.74)	1.66* (0.96, 2.87)	
M high, W low	1.27 (0.72, 2.21)	1.24 (0.71, 2.15)	1.25 (0.72, 2.19)	1.26 (0.72, 2.22)	
Both high	1.42 (0.95, 2.13)	1.46* (0.98, 2.20)	1.47* (0.98, 2.20)	1.52* (1.01, 2.29)	
Parity ¹					
Zero	1.79** (1.20, 2.68)	1.77** (1.18, 2.65)	1.92*** (1.28, 2.88)	2.05*** (1.36, 3.09)	
One	2.67*** (1.82, 3.94)	2.63*** (1.79, 3.88)	2.80*** (1.90, 4.12)	2.97*** (2.00, 4.40)	
Two or more (ref.)	1.00	1.00	1.00	1.00	
Type of relationship					
De facto	0.77 (0.60,1.11)	0.74 (0.54, 1.01)	0.74* (0.55, 1.01)	$0.65^{**}(0.47, 0.90)$	
Married (ref.)	1.00	1.00	1.00	1.00	
Controls					
Highest level of education					
Less than university (ref.)			1.00	1.00	
University			0.53*** (0.37,0.77)	0.62* (0.43,0.90)	
Country of birth					
Both Australians (ref.)				1.00	
Only M born overseas				0.70 (0.33,1.49)	
Only W born overseas				0.22** (0.09,0.58)	
Both born overseas				0.30*** (0.15,0.59)	
Lite satisfaction		1.00		1.00	
Low or normal (ref.)		1.00		1.00	
Hıgh		0.56*** (0.40,0.78)		0.50*** (0.36,0.71)	
AIC	1412	1403	1401	1372	
BIC	1499	1495	1493	1486	
Mc Fadden's R-square	0.065	0.062	0.073	0.098	
Mc Fadden's R-square Adj	0.043	0.0749	0.050	0.070	

 Table 2. Odds ratios and 95% confidence intervals assessing associations between selected characteristics and perceived infertility.

Note: Standard errors in parentheses. * p<0.05, ** p<0.01, *** p<0.001. ¹Based on female respondent. *Source:* Authors' calculations using data from the HILDA survey, wave 19, release 19.





Association between perceived infertility and contraceptive use

In Table 3, Model 1 shows that the perception of infertility is a strong predictor of not using contraception at the couple level (0.37). Using a predicted probability approach, it is estimated that the perception of infertility increases the probability of not using contraception from 19% to over 35%. When all control variables are added (Model 2), the association between perceived infertility and contraceptive use remains significant. As expected, there is also a strong association between childbearing desires and contraceptive use, with women's childbearing desires more influential in the decision to use contraception than men's childbearing desires (0.43 and 0.69, respectively).



Table 3. Odds ratios and 95% confidence intervals assessing associations between selected

 characteristics and contraceptive use.

	Contraceptive use (N=1,654) Odds ratios & 95% CI	
	Model 1	Model 2
Perceived infertility (PI)		
Yes	$0.37^{***}(0.28, 0.49)$	0.42*** (0.31, 0.57)
No (ref.)	1.00	1.00
Age ¹		
18-24 (ref.)		1.00
25-29		0.87 (0.55, 1.39)
30-34		0.57* (0.36, 0.93)
35-39		0.49** (0.29, 0.82)
40-44		0.32*** (0.18,0.57)
Desire for children		
Both low (ref.)		1.00
W high, M low		0.43*** (0.27, 0.70)
M high, W low		0.69*** (0.41, 1.17)
Both high		0.43*** (0.29, 0.64)
Short-term intention to have		
children ²		
Neither (ref.)		1.00
W only		0.39*** (0.22, 0.69)
M only		$0.42^{**}(0.25, 0.73)$
Both		0.19*** (0.13, 0.28)
Parity ¹		
Zero		1.22 (0.85, 1.76)
One		1.41 (0.97, 2.05)
Two or more (ref.)		1.00
Type of relationship		
De facto		1.23 (0.92, 1.63)
Married (ref.)		1.00
Highest level of education		
Less than university (ref.)		1.00
University		1.47* (1.08,2.00)
AIC	1763	1620
BIC	1774	1706
Mc Fadden's R-square	0.026	0.121
Mc Fadden's R-square Adjusted	0.024	0.103

Note: Standard errors in parentheses.* p<0.05, ** p<0.01, *** p<0.001.

¹Based on female respondent.

²Intention to have a child in the next 12 months.

Source: Authors' calculations using data from the HILDA survey, wave 19, release 19.





Perceived infertility and contraceptive use sub-groups

On a broad level, 66.7% (1,103 couples) of the analytic sample reported not perceiving infertility and using contraception, which was the most common subgroup. The other groups consisted of 16.9% (280) of couples that did not perceive infertility and did not use contraception, 9.7% (161) that perceived infertility and used contraception, with the remaining 6.7% (110) of the sample reporting that they perceived infertility and did not use contraception. The full table of coefficients with the base category of not perceiving infertility and using contraception is in the Appendix Table A1. For ease of interpretation, Figure 3 shows the predicted probabilities for a selected group of variables obtained from the multinomial regression model comparing perceived infertility and contraceptive use subgroups, holding all the other variables at their averages. The full table of predicted probabilities is in the Appendix Table A2. With age there is an increase in the probability of perceiving infertility. For instance, 14% of couples in which the female partner is aged 40-44 perceive infertility and do not use contraception, compared to only 2% at age 18-24. Another 10% of couples at age 40-44 perceives infertility and uses contraception compared to 7% at age 18-24. The probability of not using contraception increases with age even in the absence of a perception of infertility, from 13% in the youngest age group to over 20% in the oldest age group. Perceived health status is another key driver of the perception of infertility and of using contraception. The probability of belonging to the most common group of couples that does not perceive infertility and uses contraception markedly declines if the female partner or both partners perceive to be in poor health, from 73% to 61% and 48% respectively, while the probability of belonging to the two groups of couples with positive infertility perceptions increases, from 7% to 22% for the group that perceives infertility and uses contraception, and from 4% to 16% for the group that perceives infertility and does not use contraception. As expected, a stronger desire for children increases the predicted probability of not using contraception from 10% to 19%, but also of perceiving infertility and not using contraception, from 2% to 7%. This indicates that the desire for children affects contraceptive use in two different ways: directly and indirectly through the perception of infertility. Couples with one or no children have a higher predicted probability of identifying as infertile compared to couples with two or more children.





Figure 3. Stacked bar plots for the predicted probability of selected explanatory variables, Australia, 2019.

Source: Authors' calculations using data from the HILDA survey, wave 19, release 19.

7. Discussion

While prior studies are hampered by the use of non-representative samples and by paying only limited attention to the male partner, this study has gone beyond the state-of-the-art by being the first to comprehensively analyse the correlates of perceived infertility among a nationally representative sample of couples and to demonstrate a significant correlation between perceived infertility and contraceptive use. Taken together, the findings suggest that the perception of infertility is only partly driven by the actual biological inability to conceive and that life-course interference factors are also among its drivers. Although consistent with previous research at the individual level showing that the recognition of an infertility problem depends on the degree to which it disrupts personal plans (White *et al.* 2006), the present study has also unveiled new insights into how self-perceptions regarding infertility





are formed within the couple and how they are associated with contraceptive use. It was demonstrated that the correlates of perceived infertility are gendered, as the characteristics of the female's partner appear to be more influential than those of the male's partner in determining a perception of infertility at the couple level. Additionally, evidence was found on how the perception of infertility is a relevant reason why couples that do not wish to have a child do not use contraception.

More specifically, consistently with Hypothesis 1, age was an important factor affecting the perception of infertility, and couples in which the female partner was older than the male partner were almost twice as likely to perceive infertility compared to couples where the male partner was older. This indicates that the perception of infertility is more affected by a woman's age, which is consistent with the fact that fecundity starts declining at a younger age among women (Liu & Case 2011; Schmidt et al. 2014). Only partial support was found for Hypothesis 2. When both partners perceive to be in poor health status, the predicted probability of perceiving infertility increases from 13% to 40%. However, when only one partner is in poor health, there is a significant positive effect on the perception of infertility only if it is the female partner. This suggests that infertility may be wrongly thought of as mainly a woman's condition and highlights how the medical focus on women's body may lead to a marginalisation of men (Carmeli & Birenbaum-Carmeli 1994). Furthermore, while perceived infertility might be an important marker of overall health among women because of its strong association with self-rated health status, this might not be the case among men. In line with Hypothesis 3, couples in which both partners or the female partner only had a strong desire for children were predicted to be more likely to perceive infertility compared to couples with a low childbearing desire, 11% versus 16-17%. These results support the findings from an earlier study that suggested that childbearing desires are positively associated with the perception of infertility (Shreffler et al. 2016) and add to the existing literature by showing that if partners disagree about wishing a child, women's childbearing desires are more influential that their partners' childbearing desires. Partial support was found for Hypothesis 4. Childless couples and couples with only one child were predicted to be more likely to perceive infertility compared to couples with two or more children, which is in line with the predominance of the two-child family norm in Australia (Kippen, Evans & Gray 2007). The relationship between parity and perceived infertility was not linear: couples with only one child were more likely to perceive infertility than childless couples, 22% versus 17%. Hypothesis 5 received support, as cohabiting couples were less likely to perceive infertility compared to married couples, suggesting that infertility interferes more with lifecourse plans if the couple is married. This can be explained by the fact that, despite an increasing detachment of childbearing from formal marriage in Australia (Carmichael and McDonald 2003), the





majority of childbearing still happens within marriage (Australian Bureau of Statistics 2017). In line with Hypothesis 6, the perception of infertility is associated with a significantly higher predicted probability of not using contraception. This positive association remains even after controlling for the fundamental differences in the couples' childbearing desires and short-term intentions to have a child. Since even among couples that experience infertility, the probability of conception is not null (Osmanagaoglu *et al.* 2002), this finding supports previous research suggesting a link between perceived infertility and unwanted pregnancies (Polis & Zabin 2012, Gemmill 2018).

Despite its merits, this study does not come without limitations. First, the measurement of perceived infertility may underestimate the number of couples perceiving to be infertile. This is due to the fact that the infertility question in the HILDA survey specifically asks whether the perception of the respondent is based on medical advice, and it might exclude individuals that perceive to be infertile but that did not consult a physician. Second, although a positive association was found between perceived infertility and childbearing desires, the cross-national nature of this analysis does not allow to assess the causal direction of this relationship. Indeed, while the desire for children may have increased couples' awareness about their infertility status, it is also possible that couples experience a heightened desire for children if conception does not occur (Johnsons *et al.* 2018).

Research on infertility has primarily focused on women, as they have been traditionally considered to be more concerned about reproduction (Goldscheider & Kaufman 1996). However, the understanding of how infertility perceptions are formed and how they affect reproductive behaviour clearly calls to investigate the characteristics of both partners, due to the shared nature of reproduction. As couples keep postponing childbearing until later in life, a growing proportion of them will likely experience infertility. Hence, the understanding of how infertility perceptions are formed and of their association with biological and life-course factors will become ever more essential in fertility and family research. This study has made an important first step in this direction by accounting for the interplay between partners' characteristics in the determination of infertility perceptions and by showing how such perceptions are a key predictor of contraceptive use.





References

- Australian Bureau of Statistics (ABS). (2017). Births, Australia, 2016. Catalogue no. 3301.0, Canberra: ABS.
- Barber, J.S., Miller, W., Kusunoki, Y., Hayford, S.R., & Guzzo, K.B. (2019). Intimate relationship dynamics and changing desire for pregnancy among young women. *Perspectives on Sexual and Reproductive Health*, 51(3): 143-152. DOI: 10.1363/psrh.12119
- Brugh, V.M., & Lipshultz, L.I. (2004). Male factor infertility: Evaluation and management. *Medical Clinics* of North America, 88(2): 367-385. DOI: 10.1016/S0025-7125(03)00150-0
- Buhr, P., & Huinink, J. (2014). Fertility analysis from a life course perspective. *Advances in Life Course Research*, 21: 1-9. DOI: 10.1016/j.alcr.2014.04.001
- Bunting, L., Tsibulsky, I., & Boivin, J. (2013). Fertility knowledge and beliefs about fertility treatment:
 findings from the International Fertility Decision-making Study. Human Reproduction, 28(2):
 385-397. DOI: 10.1093/humrep/des402
- Carmeli, Y.S., & Birenbaum-Carmeli, D. (1994). The predicament of masculinity: Towards understanding of the male's experience of infertility treatments. *Sex Roles*, 30: 663-677. DOI: 10.1007/BF01544669
- Carmichael, G.A., & McDonald, P. (2003). Fertility trends and differentials. In S-E. Khoo & P. McDonald (Eds.), The transformation of Australia's Population 1970-2030 (pp. 40-76). Sydney: UNSW Press.
- Cohen, J. (1983). The cost of dichotomization. *Applied Psychological Measurement*, 7, 247-253. DOI:10.1177/014662168300700301
- Elder, G.H. Jr. (1994). Time, human agency, and social change: Perspectives on the life course. *Social Psychology Quarterly*, 51(1): 4-15. DOI: doi.org/10.2307/2786971
- Frohwirth, L., Moore, A.M., & Maniaci, R. (2013). Perceptions of susceptibility to pregnancy among U.S.
 women obtaining abortions. *Social Science and Medicine*, 99: 18–26.
 DOI:10.1016/j.socscimed.2013.10.010
- Frost, J.J., Singh, S., & Finer L.B. (2007). Factors associated with contraceptive use and nonuse, United States, 2004. *Perspectives on Sexual and Reproductive Health*, 39(2), 90–99. DOI: 10.1363/3909007
- Gemmill, A. & Cowan, S.K. (2021). Low perceived susceptibility to pregnancy as a reason for contraceptive nonuse among women with unintended births. *Demographic Research*, 44(31): 659-774. DOI: 10.4054/DemRes.2021.44.31





- Gemmill, A. (2018). Perceived subfecundity and contraceptive use among young adult US women. *Perspectives on Sexual and Reproductive Health*, 50(3): 119–127. DOI:10.1363/psrh.12072
- Gemmill, A., Sedlander, E., & Bornstein, M. (2020). Variation in self-perceived fecundity among young adult US women. *Women's Health Issues*, 31(1): 31–39. DOI:10.1016/j.whi.2020.07.002
- Gemmill, A., Sedlander, E., & Bornstein, M. (2020). Variation in self-perceived fecundity among young adult U.S. women. *Women's Health*, 31(1): 31-39. DOI: 10.1016/j.whi.2020.07.002
- Goldscheider, F.K., & Kaufman, G. (1996). Fertility and commitment: Bringing men back in. Population and Development Review, 22: 87-99. DOI: 10.2307/2808006
- Greil, A.L., Leitko, T.A., & Porter, K.L. (1988). Infertility: His and Hers. *Gender & Society*, 2(2): 172-199. DOI: 10.1177/089124388002002004
- Greil, A.L., Leyser-Whalen, O., McQuillan, J., Shreffler, K.M., Lowry, M., & Johnson, K.M. (2014). Patterns of seeing a doctor and self-labelling as infertile: A longitudinal analysis of a representative sample of US women. *Sociology of Health & Illness*, 32: 140-162. DOI: 10.1177/0192513X17699027
- Greil, A.L., McQuillan, J., Johnson, K., Slauson, K., & Shreffler, K.M. (2010). The hidden infertile: Infertile women without pregnancy intent in the United States. *Fertility and Sterility*, 93(6): 2080-2083.
 DOI: 10.1016/j.fertnstert.2009.08.024
- Greil, A.L., Shreffler, K.M., Johnson, K.M., McQuillan, J., & Slauson-Blevins, K. (2013). The importance of social cues for discretionary health services utilization: The case of infertility. *Sociological Inquiry*, 83(2): 209-237. DOI: 10.1111/soin.12000.
- Hjelmstedt, A., Andersson, L., Skoog-Svanberg, A., Bergh, T., Boivin, J., & Collins, A. (1999). Gender differences in psychological reactions to infertility among couples seeking IVF- and ICSItreatment. *Comparative Studies*, 78(1): 42-8.
- Isidori, A.M., Pozza, C., Gianfrilli, D., & Isidori, A. (2006). Medical treatment to improve sperm quality. *Reproductive Biomedicine Online*, 12(6): 704-714. DOI: 10.1016/s1472-6483(10)61082-6
- Johnson, K.M., Greil, A.L., Shreffler, K.M., & McQuillan, J. (2018). Fertility and infertility: Toward an integrative research agenda. *Population Research and Policy Review*, 37: 641-666. DOI: 10.1007/s11113-018-9476-2
- Johnson, K.M., Greil, A.L., McQuillan, J., Leyser-Whalen, O., & Shreffler, K.M. (2020). The infertility and self-identification: The indeterminacy of the illness-self relationship. *Sociological Perspectives*, 63(3): 388-405. DOI: 10.1177/0731121419867691
- Kelly-Weeder, S., & Cox, C.L. (2006). The impact of lifestyle risk factors on female infertility. *Women & Health*, \$\$9\$): 1-23. DOI: 10.1300/j013v44n04_01





- Kippen, R., Evans, A., & Gray, E. (2007). Parental preference for sons and daughters in a western industrial setting: evidence and implications. *Journal of Biosocial Science*, 39(4): 583-597. DOI: 10.1017/S0021932006001477
- Klemetti, R., Raitanen, J., Sihvo, S., Saarni, S., & Koponen, P. (2010). Infertility, mental disorder and wellbeing – a nationwide survey. Acta obstetrician et gynecologica Scandinavica, 89(5): 677-682. DOI: 10.3109/00016341003623746
- Lazzari, E. (2021). Pathways into childbearing delay of men and women in Australia, *Longitudinal and Life Course Studies*. DOI: 10.1332/175795921X16197735939121
- Leridon, H. (2004). Can assisted reproduction compensate for the natural decline in fertility with age? A model assessment. *Human Reproduction*, 19(7): 1548-1553. DOI: 10.1093/humrep/deh304
- Liu, K., & Case, A. (2011). Advanced reproductive age and fertility. *Journal of Obstetrics and Gynaecology Canada*, 33(11): 1165-1175. DOI: 10.1016/S1701-2163(16)35087-3
- Loftus, J. (2009). "Oh, no, I'm not infertile": Culture, support groups, and the infertile identity. *Sociological Focus*, 42(4): 394-416. DOI: 10.1080/00380237.2009.10571364
- Luk, B. H., & Loke A.Y. (2015). The impact of infertility on the psychological well-being, marital relationships, sexual relationships, and quality of life of couples: A systematic review. *Journal of sex & marital therapy*, 41(6): 610-625. DOI: 10.1080/0092623X.2014.958789
- McDonald, P., & Reimondos, A. (2013). Relationship pathways and first birth in Australia, in A. Evans and J. Baxter (eds) *Negotiating the Life Course: Stability and Change in Life Pathways*, Dordrecht: Springer, pp. 69-98.
- McQuillan, J, Greil, A.L., White, L., & Jacob, M.C. (2003). Frustrated infertility: Infertility and psychological distress among women. *Journal of Marriage and Family*, 65: 1007-1018. DOI: 10.1111/j.1741-3737.2003.01007.x
- McQuillan, J., Torres Stone, R.A., & Greil, A.L. (2007). Infertility and life satisfaction among women. Journal of Family Issues, 28(7): 955-981. DOI: 10.1177/0192513X07300710
- Mechanic, D. (1968). Medical sociology: a selective view. New York: Free Press.
- Osmanagaoglu, K., Collins, J.A., Kolibianakis, E., Tournaye, H., Camus, H., Van Steirteghem, A., & Devroey, P. (2002). Spontaneous pregnancies in couples who discontinued intracytoplasmic sperm injection treatment: a 5-year follow-up study. *Fertility and Sterility*, 78(3): 550-556. DOI: 10.1016/S0015-0282(02)03300-9
- Passet-Wittig, J., & Greil, A. (2021). Factors associated with medical help-seeking for infertility in developed countries: A narrative review of recent literature. *Social Science & Medicine*, 277: 113782. DOI: 10.1016/j.socscimed.2021.113782





- Passet-Wittig, J., McQuillan, J., Bujard, M., & Greil, A. (2020). Is perception of inability to procreate a temporal phenomenon? A longitudinal exploration of changes and determinants among women and men of reproductive age in Germany. *Advances in Life Course Research*, 45. DOI: 10.1016/j.alcr.2020.100339
- Peterson, B.D., Newton, C.R., & Rosen, K.H. (2003). Examining congruence between partners' perceived infertility-related stress and its relationship to marital adjustment and depression in infertile couples. *Family Process*, 42(1): 59-70. DOI: 10.1111/j.1545-5300.2003.00059.x
- Polis, C.B. & Zabin, L.S. (2012). Missed conceptions or misconceptions: perceived infertility among unmarried young adults in the United States. *Perspectives on Sexual and Reproductive Health*, 44(1): 30–38. DOI: 10.1363/4403012
- Polis, C.B., Otupiri, E., Hindin, M., Chiu, D.W., Keogh, S.C., Aidoo, C., Larsen-Reindorf, R., & Bell, S.O. (2020). Prevalence and correlates of perceived infertility in Ghana. *Studies in Family Planning*, 51(3): 207-224. DOI: 10.1111/sifp.12136
- Rendall, MS, Clarke, L., Peters, H.E., Rankit, N., & Verropoulou, G. (1999). Incomplete reporting of men's fertility in the United States and Britain: a research note. *Demography*, 36(1): 135-44. DOI: 10.2307/2648139
- Richters, J., Fitzadam, S., Yeung, A., Caruana, T., Rissel, C., Simpson, J.M., & de Visser, R.O. (2016). Contraceptive practices among women: the second Australian study of health and relationships. *Contraception*, 94(5): 548-555. DOI: 10.1016/j.contraception.2016.06.016
- Schimmack, U., Oishi, S., Furr, R.M., Funder, D.C. (2004). Personality and Life Satisfaction: A Facet-Level Analysis. *Personality and Social Psychology Bulletin*. 30(8):1062-1075.
 DOI:10.1177/0146167204264292
- Schmidt, L., Sobotka, T., Bentzen, J.G., Nyboe Andersen, A., & ESHRE Reproduction and Society Task Force, (2012). Demographic and medical consequences of the postponement of parenthood. *Human Reproduction Update*, 18(1): 29-43.
- Schreffler, K.M., Tiemeyer, S., Dorius, C., Spierling, T., Greil, A., & McQuillan, J. (2016). Infertility and fertility intentions, desires, and outcomes among US women. *Demographic Research*, 35 (39): 1149-1168. DOI: 10.4054/DemRes.2016.35.39
- Settersten, R.A. (2015). Relationships in time and the life course: The significance of linked lives. *Research in Human Development,* 12(3-4): 217-223. DOI: 10.1080/15427609.2015.1071944
- Slauson-Blevins, K., McQuillan, J., & Greil, A.L. (2013). Online and in-person health-seeking for infertility. Social Science & Medicine, 99: 110-115.





- Taft, A.J., Shankar, M., Black, K.I., Mazza, D., Hussainy, S., & Lucke, J.C. (2018). Unintended and unwanted pregnancy in Australia: A cross-sectional, national random telephone survey of prevalence and outcomes. *Medical Journal of Australia*, 209 (9): 407-408. DOI: 10.5694/mja17.01094
- Testa, M. R., & Bolano, D. (2021). When partners disagreement prevents childbearing: A couple-level analysis in Australia. *Demographic Research*, 44(33): 811-838. DOI: 10.4054/DemRes.2021.44.33
- Wagner, M., Huinink, J. & Liefbroer, A. (2019). Running out of time? Understanding the consequences of the biological clock for the dynamics of fertility intentions and union formation. *Demographic Research*, 40(10): 1-26. DOI: 10.4054/DemRes.2019.40.1
- Watson, N., & Wooden, M. (2010). The HILDA Survey: A case study in the design and development of a successful household panel study. *Longitudinal and Life Course Studies*, 3(3): 369-381. DOI: 10.14301/llcs.v3i3.208
- Weitzman, A., Barber, J.S., Kusunoki, Y., & England, P. (2017). Desire for and to Avoid Pregnancy During the Transition to Adulthood. *Journal of Marriage and Family*, 79(4): 1060-1075. DOI: 10.1111/jomf.12396
- White, L., McQuillan, J., Greil, A.L., & Johnson, D.R. (2006). Infertility: Testing a helpseeking model. *Social Science & Medicine*, 62: 1031-1041. DOI: 10.1016/j.socscimed.2005.11.012
- Wilson, E.K., & Koo, H.P. (2006). The relationship context: Its effects on low-income women's desire for a baby. *Journal of Marriage and Family*, 68(5): 1326-1340. DOI: 10.1111/j.1741-3737.2006.00331.x
- Yano, J.C., Lundsberg, L.S., & Pal, L. (2014). Racial differences in fertility knowledge and awareness amongst reproductive age women in the U.S. *Fertility and Sterility*, 102 (30: 241-242.
- Zegers-Hochschil, F., Adamson, G.D., Dyer, S., Racowsky, C., de Mouzon, J., Sokol, R., Rienzi, L., Sunde, A., Schmidt, L., Cooke, I.D., Simpson, J.L., & van der Poel, S. (2017). The international glossary of infertility and fertility care, 2017. *Human Reproduction*, 32(9): 1786-1801. DOI: 10.1016/j.fertnstert.2017.06.005
- Zola, I.K. (1973). Pathways from the doctor: from person to patient. *Social Science & Medicine*, 7: 677-689.



Appendix

Table A1. Descriptive statistics.

Name	Mean %	Description
Perceive infertility	16.6	Measured with the question: Based on medical advice, do you know of
		any physical or health reason that would make it difficult for you (and/or
		your partner) to have [children / more children]? Coded as 0 if both
		partners did not perceive infertility, 1 if at least one partner perceived
		infertility.
Use contraception	75.4	Measured with the question: Do you (and your partner) use birth control
		measures? Coded as 0 if both partners did not use contraception, 1 if at
		least one partner used contraception.
Age		Coded as 0 if the female partner was between 18 and 24 years of age.
18-24	13.6	
25-29	24.8	
30-34	26.1	
35-39	19.2	
40-44	16.2	
Perceived health status		Perceived health status was measured with the question: In general,
Both good	82.3	would you say your health is: [1] excellent, [2] very good, [3] good, [4]
Woman good, man poor	7.5	fair, [5] poor. Coded as 0 if both partners believe to have excellent, very
Man good, woman poor	8.2	good, or good health.
Both poor	2.0	
Parity		The total number of children ever born were used to construct parity.
Zero	37.0	Coded as 0 if the female partner does not have any children.
One	18.4	
Two or more	44.7	
Desire for children		Measured with the question: I now want you to pick a number between
Both low	31.4	0 and 10 to show how you feel about having (more) children /a child in
Woman high, man low	7.9	the future. [The more definite you are that you would like to have (more)
Man high, woman low	8.3	children /a child, the higher the number you should pick]. Coded as 0 if
Both high	52.4	neither partner gave a score higher than 4.
¢Type of relationship		Coded as 0 if the members of the couple were in a cohabiting
Cohabitation	43.6	relationship or coded as 1 if the members of the couple were legally
Marriage	56.4	married.
Short-term childbearing		Intentions to have children were measured with the questions: How
intentions		many more children do you intend to have (including zero) and In which
Neither	81.8	year do you intend to have a / your next child? Coded as 0 if neither
Woman only	3.7	partner intend to have a child/ their next child within the next 12
Man only	4.3	months.
Both	10.2	
Life satisfaction		Measured with the question: All things considered, how satisfied are
Low	18.2	you with your life? Pick a number between 0 and 10 to indicate how
High	81.8	satisfied you are. Coded as 0 if neither partner gave a score higher than
		7.
Highest level of education		Coded as 0 if the highest educational attainment of at least one
Less than university	75.9	member of the couple was below tertiary.
University	24.1	
Country of birth		Coded as 0 if both partners were born either in Australia or in another
Both Australians	82.6	English speaking country.
Woman only Australian	5.1	
Man only Australian	3.9	
Both born overseas	8.3	





Table A2. Multinomial logistic regression results comparing perceived infertility and contraceptive use subgroups.

	Do not perceive		Perceive infertility and
	infertility and do not use	Perceive infertility and	do not use contraception
	contraception	use contraception	(N=110)
	(N=280)	(N=161)	Odds ratios & 95% CI
	Odds ratios & 95% CI	Odds ratios & 95% CI	
Desire for children			
Both low (ref.)	1.00	1.00	1.00
W high, M low	2.33** (1.36, 4.01)	1.48 (0.76, 2.89)	3.23** (1.36, 7.68)
M high, W low	1.44 (0.80, 2.60)	1.22 (0.63, 2.36)	1.62 (0.60, 4.37)
Both high	2.33*** (1.48, 3.65)	1.13 (0.67, 1.89)	3.32*** (1.60, 6.88)
Parity ¹			
Zero	0.63* (0.42, 0.96)	1.50 (0.89, 2.53)	2.29** (1.23, 4.29)
One	0.57* (0.36, 0.88)	2.47*** (1.50, 4.05)	2.67** (1.45, 4.90)
Two or more (ref.)	1.00	1.00	1.00
Type of relationship			
De facto	0.96 (0.69, 1.34)	0.78 (0.52, 1.16)	0.53** (0.32, 0.86)
Married (ref.)	1.00	1.00	1.00
Perceived health status			
Both good (ref.)	1.00	1.00	1.00
W good, M poor	1.64 (1.00, 2.69)	0.87 (0.43, 1.74)	1.16 (0.52, 2.56)
W poor, M good	1.27 (0.74, 2.19)	2.81*** (1.71, 4.63)	1.69 (0.83, 3.46)
Both poor	1.35 (0.41, 4.42)	4.39*** (1.78, 10.81)	5.50** (1.78, 16.99)
Short-term intention to have			
children ²			
Neither (ref.)	1.00	1.00	1.00
W only	2.95*** (1.57, 5.54)	1.12 (0.43, 2.93)	2.04 (0.76, 5.46)
M only	2.15* (1.15, 4.01)	0.58 (0.20, 1.71)	2.20 (0.95, 5.07)
Both	6.18*** (3.93, 9.71)	1.17 (0.58, 2.38)	4.77*** (2.64, 8.63)
Age ¹			
18-24 (ref.)	1.00	1.00	1.00
25-29	0.96 (0.57,1.62)	1.23 (0.69, 2.20)	2.00 (0.81, 4.93)
30-34	1.46 (0.85, 2.49)	1.11 (0.59, 2.12)	2.70* (1.07, 6.79)
35-39	1.50 (0.82, 2.75)	1.54 (0.75, 3.17)	5.15*** (1.94, 13.69)
40-44	2.21* (1.15, 4.24)	2.02 (0.94, 4.34)	11.21*** (3.96, 31.75)
Age difference			
No difference (ref.)	1.00	1.00	1.00
W more than 3 years older	1.27 (0.93, 1.73)	1.06 (0.71, 1.59)	1.44 (0.90, 2.31)
M more than 3 years older	1.11 (0.57, 2.14)	1.71 (0.81, 3.59)	2.05 (0.93, 4.51)
Highest level of education	1.00	1.00	1.00
Less than university (ref.)	1.00	1.00	1.00
University	0.58** (0.41, 0.84)	0.61* (0.38, 0.99)	0.50** (0.29, 0.87)
Country of birth	1.00	1.00	1.00
Both Australians (ref.)	1.00	1.00	1.00
Only M born overseas	1.33 (0.65, 2.72)	0.40(0.12, 1.35)	1.41 (0.55, 3.62)
Unly W born overseas	2.14** (1.24, 3.68)	$0.10^* (0.01, 0.73)$	0.52 (0.17, 1.56)
Both born overseas	2.3/*** (1.47, 3.80)	0.22** (0.08, 0.64)	0.61 (0.25, 1.49)
Life satisfaction	1.00	1.00	1.00
Low or normal (ref.)	1.00	1.00	1.00
High	1.20 (0.81, 1.79)	0.46^{***} (0.30, 0.69)	0.67(0.39, 1.15)

Note: The base category is "Do not perceive infertility and use contraception". Standard errors in parentheses.* p<0.05, ** p<0.01, *** p<0.001. ¹Based on female respondent. ²Intention to have a child in the next 12 months.

Source: Authors' calculations using data from the HILDA survey, wave 19, release 19.





Table A3. Predicted probabilities (95% CI) of perceived infertility and contraceptive use subgroups (N= 1,654).

	Do not perceive	Do not perceive	Perceive infertility	Perceive infertility
	infertility and do	infertility and do	and use	and do not use
	use contraception	not use	contraception	contraception
		contraception		
Age ¹				
18-24	79 (73,85)	13 (8,18)	7 (3,10)	2 (0,3)
25-29	77 (72,81)	12 (9,15)	8 (5,11)	3 (2,5)
30-34	72 (67,76)	17 (13,21)	7 (4,9)	4 (2,6)
35-39	67 (61,73)	16 (12,21)	9 (5,12)	7 (4,11)
40-44	56 (49,64)	20 (14,26)	10 (5,14)	14 (8,20)
Age difference				
No difference	73 (70,76)	15 (12,17)	8 (6,10)	4 (3,6)
M more than 3 years older	68 (63,73)	18 (14,21)	8 (5,10)	6 (4,8)
F more than 3 years older	65 (54,76)	15 (7,22)	12 (5,19)	8 (2,13)
Perceived health status				
Both good	73 (70,76)	15 (13,17)	7 (6,9)	4 (3,6)
W good, M poor	67 (58,76)	22 (14,30)	6 (2,10)	5 (1,8)
W poor, M good	61 (52,69)	16 (9,22)	17 (11,24)	6 (2,10)
Both poor	48 (30,67)	13 (1,26)	22 (8,35)	16 (3,30)
Desire for children				
Both low	79 (75,84)	10 (7,13)	8 (5,11)	2 (1,4)
W high, M low	65 (56,73)	13 (7,19)	9 (4,14)	4 (1,7)
M high, W low	74 (66,81)	19 (12,26)	10 (4,15)	6 (2,11)
Both high	66 (62,70)	19 (16,23)	7 (5,10)	7 (5,9)
Parity ¹				
Zero	71 (67,76)	13 (10,16)	9 (6,12)	7 (4,9)
One	68 (62,73)	11 (7,14)	14 (8,18)	7 (4,10)
Two or more	71 (67,75)	20 (17,24)	6 (4,8)	3 (2,4)
Type of relationship				
De facto	74 (70,77)	16 (13,19)	7 (5,9)	3 (2,5)
Married	70 (66,73)	15 (13,18)	9 (7,11)	6 (4,8)
Short-term intention to have				
children ²				
Neither	75 (72,77)	10 (7,13)	8 (5,11)	2 (1,4)
W only	65 (56,73)	19 (12,26)	10 (4,15)	7 (2,11)
M only	74 (66,81)	13 (7,19)	9 (4,14)	4 (1,7)
Both	66 (62,70)	19 (16,23)	7 (5,10)	7 (5,9)
Life satisfaction				
Low or normal	67 (61,73)	12 (9,16)	14 (10,19)	6 (3,9)
High	72 (69,75)	16 (14,18)	7 (5,9)	4 (3,6)
Highest level of education				
Less than university	69 (66,72)	17 (15,19)	9 (7,11)	5 (4,7)
University	79 (75,83)	11 (8,15)	6 (4,9)	3 (2,5)
Country of birth				
Both Australians	71 (68.73)	14 (12,16)	11 (9,12)	2 (0,4)
Only M born overseas	71 (59.82)	18 (8,28)	2 (0,4)	7 (1.13)
Only W born overseas	68 (58,79)	28 (18,34)	1 (0,3)	2 (0,5)
Both born overseas	65 (56,74)	30 (21,39)	2 (0,4)	3 (0,5)

¹Based on female respondent.

²Intention to have a child in the next 12 months.

Source: Authors' calculations using data from the HILDA survey, wave 19, release 19.